Author’s response to reviews

Title: Mechanisms of ring chromosome formation, ring instability and clinical consequences.

Authors:

Roberta S Guilherme (robertaguilherme@hotmail.com)
Vera F Ayres Meloni (vera.meloni@gmail.com)
Chong A Kim (chong.kim@icr.usp.br)
Renata Pellegrino (renata.pellegrino@gmail.com)
Sylvia S Takeno (sylvinha66@yahoo.com.br)
Nancy B Spinner (spinner@mail.med.upenn.edu)
Laura K Conlin (conlinl@email.chop.edu)
Denise M Christofolini (denise.morf@gmail.com)
Leslie D Kulikowski (lesliekulik@gmail.com)
Maria I Melaragno (melaragno.morf@epm.br)

Version: 7 Date: 20 December 2011

Author’s response to reviews:

We wrote a clear statement of consent for the publication of the images in Figure 1 in the manuscript (e.g. “Written informed consent was obtained from the patients for publication of this report and accompanying images.”).

We removed the Key words from the Abstract.

We rearranged these sections in the following order: Competing Interests, Authors’ Contributions, Acknowledgements.

We read and checked the manuscript again.